**Author’s response to reviews**

**Title:** The Effect of Education around Ethical Principles on Nurses’ Perception to Patient Safety Culture in an Iranian Psychiatric Unit: A Pilot Study

**Authors:**

Behzad Razzani (razzani.behzad@yahoo.com)

Foroozan Atashzadeh –Shoorideh (f_atashzadeh@sbmu.ac.ir)

Tayebeh Jamshidi (tayebe.jamshidi@yahoo.com)

Maasoumeh Barkhordari-Sharifabad (barkhordary.m@gmail.com)

Zahra Lotfi (zlotfi1024@gmail.com)

Victoria Skerrett (Victoria.Skerrett@bcu.ac.uk)

**Version:** 3 **Date:** 27 Jan 2020

**Author’s response to reviews:**

Thank you very much for your suggestions for improving this manuscript.

Following your comments, the manuscript edited by a native English speaking colleague and The requested corrections were made.

In the Ethics approval and the consent to participate section, it added the name of the institution.

The authors reiterate their gratitude to the editor and the reviewers who helped with valuable comments to improve the clarification and quality of this study. Thank you for the opportunity to make changes and submit a revised paper.